Author's response to reviews

Title: The spectrum of Apert syndrome: phenotype, particularities in orthodontic treatment, and characteristics of orthognathic surgery

Authors:

Ariane Hohoff (hohoffa@uni-muenster.de)
Ulrich Joos (joos@uni-muenster.de)
Ulrich Meyer (ulrich.meyer@med.uni-duesseldorf.de)
Ulrike Ehmer (ehmer@uni-muenster.de)
Thomas Stamm (stammt@uni-muenster.de)

Version: 2 Date: 18 January 2007

Author's response to reviews: see over
Letter to the reviewers,

thank you for the constructive suggestions to enhance the clarity of my paper. I followed your suggestions and removed the original figure 14. I do hope that the article is ready for publishing now.

Thank you.

A. Hohoff